Author’s response to reviews

Title: Cervical cancer screening service utilization and associated factors among HIV positive women attending adult ART clinic in Public Health Facilities, Hawassa town, Ethiopia: A cross-sectional study

Authors:
Abiyu Ayalew Assefa (abiyman143@gmail.com)
Feleke Hailemichael Astawesegn (felekeh86@gmail.com)
Betelhem Eshetu (betelhemeshetu123@gmail.com)

Version: 3 Date: 02 Nov 2019

Author’s response to reviews:

Thank you very much for your valuable comments to enrich the manuscript

Reviewer: In your “ethical approval and consent to participate” section, please confirm whether informed consent was obtained in written or verbal format.
Response: It is verbal consent; since it has no any procedure that have a significant risk to participant, we simply asked them about their cervical cancer screening service experience. The institutional ethical review board has approved the protocol

2. Questionnaire
Reviewer: Please clarify whether the questionnaire used in your study was developed for this study or has previously been published elsewhere. If the questionnaire has been published elsewhere please provide a reference to it in your manuscript, if the questionnaire was developed for this study please upload an English language version as a supplementary file.
RESPONSE: we adopted the questionnaire from different literatures and we have cited them in under data collection section in the manuscript (line 130)

Reviewer: However, you still need to pay attention, throughout the whole text, about :THE USE OF DECIMALS (ONE DECIMAL IS ENOUGH);and THE APPROPRIATE USE OF CCS (CERVICAL CANCER SCREENING) and CCSS (CERVICAL CANCER SCREENING SERVICE)
Response: we have made appropriate correction and it is consistent throughout the document.